Author’s response to reviews

Title: Comparison of device-based physical activity and sedentary behaviour following percutaneous coronary intervention in a cohort from Sweden and Australia: a harmonised, exploratory study.

Authors:

Nicole Freene (Nicole.Freene@canberra.edu.au)
Sabina Borg (sabina.borg@liu.se)
Margaret McManus (margaret.macmanus@act.gov.au)
Tarryn Mair (tarryn.mair@act.gov.au)
Ren Tan (ren.tan@act.gov.au)
Rachel Davey (raachel.davey@canberra.edu.au)
Birgitta Oberg (birgitta.oberg@liu.se)
Maria Back (maria.back@liu.se)

Version: 2 Date: 13 Feb 2020

Author’s response to reviews:

Dear BMC Sports Science, Medicine and Rehabilitation Editorial Team,

Thank you for considering our manuscript entitled, ‘Comparison of device-based physical activity and sedentary behaviour following percutaneous coronary intervention in a cohort from Sweden and Australia: a harmonised, exploratory study’, for publication in BMC Sports Science, Medicine and Rehabilitation.

We appreciate the comments from the Assistant Editor and have made revisions and responded to each comment.

2. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Currently the individual contributions of RT, RD and BO are not included.

The contributions of RT, RD and BO have been acknowledged in the Authors’ Contributions section (page 20, lines 332-335). All three authors contributed to the design of the study, the
interpretation of the data, contributed to the manuscript, approved the submitted versions and agreed to be personally accountable for the work.

3. Please remove any additional files you do not want published alongside your manuscript, including cover/response letters.

All additional files have been removed as requested.

4. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

The above instructions have been followed, and a clean version of the final manuscript has been uploaded.

Associate Editor Comments:

Please replace objectively measured physical activity with the more contemporary description device-based physical activity.

Thank you for this comment. Objectively measured physical activity has been replaced by device-based physical activity throughout the manuscript.

We look forward to hearing from you in due course.

Kind regards,

Nicole